Author’s response to reviews

Title: Perceived family support and condom use among secondary school female students in Limbe urban city of Cameroon Elvis E Tarkang 1 1HIV/AIDS Prevention Research Network, Cameroon (HIVPREC), Commonwealth Avenue, Opposite Premier Pharmacy, P.O Box 36, Kumba, Cameroon. Email address: ebeyang1@yahoo.com

Authors:

Elvis E Tarkang (ebeyang1@yahoo.com)

Version: 2 Date: 27 January 2013

Author’s response to reviews:

COVER LETTER
Revised manuscript: Perceived family support and condom use among secondary school female students in Limbe urban city of Cameroon.

Dear Editor,

It is with great honour and pleasure that I resubmit this manuscript. I have tried to adhere to all the BMC Public Health guidelines and requirements and hereby submit the revised version.

The following sections were revised according to BMC guidelines:
• The title
• The abstract
• The background
• The results
• The references

Thank you very much for granting me an opportunity to revise the manuscript.

Yours sincerely

Dr Elvis E Tarkang (Author)